Author's response to reviews

Title: Using qualitative mixed methods to study small health care organizations while maximizing trustworthiness and authenticity

Authors:

Christine B Phillips (christine.phillips@anu.edu.au)
Kathryn M Dwan (kathryn.dwan@anu.edu.au)
Julie Hepworth (julie.hepworth@qut.edu.au)
Christopher M Pearce (chris_pearce@mac.com)
Sally Hall (sally.hall@anu.edu.au)

Version: 4 Date: 12 September 2014

Author's response to reviews:

12 September 2014

Professor Gianfranco Damiani
Editor
BMC Health Services Research
Dear Professor Damiani

REVISION: Using qualitative mixed methods to study small health care organizations while maximizing trustworthiness and authenticity

Thank you for the opportunity to respond to the reviewers’ comments on this paper, which have been very helpful in improving it. I have attached a detailed summary of the comments in the attached table.

Sincerely

Christine Phillips
Associate Professor, Social Foundations of Medicine
ANU Medical School